

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Casi-Eberhard, Elisa; Mardan, Mohammed; Muddasani, Rohit Reddy

Working Paper So close and yet so far: The ability of mandatory disclosure rules to crack down on offshore tax evasion

WIDER Working Paper, No. 2022/116

Provided in Cooperation with: United Nations University (UNU), World Institute for Development Economics Research (WIDER)

Suggested Citation: Casi-Eberhard, Elisa; Mardan, Mohammed; Muddasani, Rohit Reddy (2022) : So close and yet so far: The ability of mandatory disclosure rules to crack down on offshore tax evasion, WIDER Working Paper, No. 2022/116, ISBN 978-92-9267-250-8, The United Nations University World Institute for Development Economics Research (UNU-WIDER), Helsinki, https://doi.org/10.35188/UNU-WIDER/2022/250-8

This Version is available at: https://hdl.handle.net/10419/273905

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



WIDER Working Paper 2022/116

So close and yet so far: the ability of mandatory disclosure rules to crack down on offshore tax evasion

Elisa Casi,¹ Mohammed Mardan,¹ and Rohit Reddy Muddasani²

October 2022

United Nations University World Institute for Development Economics Research



Abstract: We study the short-term effect of the introduction of the mandatory disclosure programme for aggressive tax arrangements by focusing on the one introduced in May 2018 under Council Directive 2018/288/EU (or DAC6). Employing bilateral data on cross-border deposits, we study the effect of this new disclosure requirement on cross-border tax evasion. Our results show a reduction of cross-border deposits in EU countries with strong enforcement, captured by large monetary penalties for misreporting. At the same time, we document a relocation of income and wealth to countries with limited intermediary reporting obligations. Finally, we detect a short-term increase of US\$14 billion in cross-border deposits held in countries offering citizenship/residence by investment programmes, suggesting the use of these schemes as regulatory arbitrage to circumvent the disclosure mandated under DAC6. We provide timely and relevant evidence contributing to the debate on international administrative cooperation to reduce cross-border tax evasion.

Key words: tax evasion, intermediary reporting, administrative cooperation, cross-border deposits

JEL classification: H26, G21, F42

Acknowledgements: We appreciate the valuable suggestions and comments from Michael Carlos Best, Eva Eberhartinger, Evelina Gavrilova-Zoutman, Niels Johannesen, Dominika Langenmayr, Raphael Mueller, Carol Seregni, Barbara Stage, Finn Tarp, Maximilian Todtenhaupt, Johannes Voget, and the audience at the Norwegian Centre for Taxation and NHH Norwegian School of Economics, Vienna University of Economics and Business, Paderborn University, Accounting Summer Camp, IIPF Annual Congress, Mannheim Taxation Annual Conference, World Bank Conference, VHB Faculty Workshop. We gratefully acknowledge support from the German Research Foundation Project-ID 403041268—TRR 266 Accounting for Transparency, from the Austrian Science Fund (FWF): Doc 92-G, and from the United Nations University World Institute for Development Economics Research (UNU-WIDER) project on illicit financial flows.

Copyright © UNU-WIDER 2022

Information and requests: publications@wider.unu.edu

ISSN 1798-7237 ISBN 978-92-9267-250-8

https://doi.org/10.35188/UNU-WIDER/2022/250-8

Typescript prepared by Gary Smith.

The Institute is funded through income from an endowment fund with additional contributions to its work programme from Finland, Sweden, and the United Kingdom as well as earmarked contributions for specific projects from a variety of donors.

Katajanokanlaituri 6 B, 00160 Helsinki, Finland

The views expressed in this paper are those of the author(s), and do not necessarily reflect the views of the Institute or the United Nations University, nor the programme/project donors.

¹NHH Norwegian School of Economics, Bergen, Norway; ²Vienna University of Economics and Business, Vienna, Austria; corresponding author: Elisa Casi, elisa.casi@nhh.no

This study has been prepared within the UNU-WIDER project Detecting and countering illicit financial flows that is implemented in collaboration with the University of Copenhagen. The project is part of the Domestic Revenue Mobilization programme, which is financed through specific contributions by the Norwegian Agency for Development Cooperation (Norad).

UNU-WIDER employs a fair use policy for reasonable reproduction of UNU-WIDER copyrighted content—such as the reproduction of a table or a figure, and/or text not exceeding 400 words—with due acknowledgement of the original source, without requiring explicit permission from the copyright holder.

United Nations University World Institute for Development Economics Research provides economic analysis and policy advice with the aim of promoting sustainable and equitable development. The Institute began operations in 1985 in Helsinki, Finland, as the first research and training centre of the United Nations University. Today it is a unique blend of think tank, research institute, and UN agency—providing a range of services from policy advice to governments as well as freely available original research.

1 Introduction

On 24 August 2022, the chair of the Senate Finance Committee, Ron Wyden, released the findings of a year-long investigation concerning allegations of what has been called 'the largest tax evasion case brought against an individual in U.S. history'. Robert Brockman, a former CEO of an Ohio-based software company, was accused of concealing approximately US\$2.7 billion in income from the IRS through the so-called 'shell bank' loophole, a scheme involving the use of offshore entities and secret bank accounts to escape the reporting duty under the Foreign Account Tax Compliance Act (FATCA) (US Senate Finance Committee 2022).

Globally, collecting tax revenues has become increasingly challenging. As the above case suggests, the pervasive use of complex legal structures to hide wealth and related income overseas and evade tax obligations at home is a great concern to policy-makers. The related revenue loss has been estimated to be approximately US\$200 billion annually (Zucman 2013), and is mainly attributed to the top 0.1 per cent highest earners, as empirical evidence shows (Alstadsæter et al. 2018; Guyton et al. 2021). Recent evidence from several leaks, such as the Panama Papers in 2016, the Paradise Papers in 2017, the Pandora Papers in 2021, and the Suisse Secrets in 2022, revealed—just as in the 'shell bank' loophole—the key role played by tax advisers, lawyers, financial institutions, and other intermediaries in supporting the world's economic elite in under-reporting income and wealth in their respective countries of residence.

In this study we investigate the effect of an innovative reporting standard, called mandatory disclosure rules (MDRs), which has the power to prevent individuals from exploiting tax evasion schemes such as the 'shell bank' loophole. Specifically, an MDR targets enablers of cross-border tax evasion by requiring them to disclose their clients' tax schemes. In contrast to existing tax transparency initiatives, such as FATCA of the United States and the Common Reporting Standard (CRS) of the OECD, the key innovation of MDRs is that it requires intermediaries, such as consultants, lawyers, or financial institutions, to report to local tax authorities a comprehensive set of information on all currently used transactions that have certain elements of tax aggressiveness.¹ In this way, tax authorities obtain information that is not limited to a specific channel of tax evasion (e.g. financial assets in the case of FACTA and the CRS), but rather extends to all types of tax-aggressive arrangements.

The United States was one of the first countries to introduce MDRs, in the 1980s. Nowadays, MDRs exist in several countries around the world.² We focus on the one introduced in the EU in June 2018 under EU Council Directive 2018/822/EU, also known as DAC6. While other countries have domestic MDRs, the EU is the first to introduce this new disclosure rule under a multilateral approach where the information collected under DAC6 is automatically exchanged across EU member states. An MDR like the one under DAC6 can enhance the speed and accuracy of the assessment of whether a transaction is created only for the purpose of evading taxes. For example, under the UK MDR more than 3,000 transactions have been reported over seven years and legislation has been changed in relation to almost 600 reported transactions (Devereux et al. 2012). Anecdotal evidence from the US MDR suggests that it was key in countering the expansion of corporate tax shelters in the early 2000s (Noked and Marcone 2022).

¹ Information to be reported includes a summary of the content of the transaction, the value of the transaction, the identification detail of the intermediaries, and of the relevant taxpayers. The reporting obligation extends to taxpayers—for example, in the case that the intermediary may claim legally recognized professional secrecy or the intermediary is not a resident of the EU or there is no intermediary involved in a transaction.

² For an overview of existing MDRs, see Noked et al. (2022).

Similarly, EU tax authorities can largely benefit from the information collected under DAC6. Assume, for example, a scheme similar to the one used in the 'shell bank' case: a resident of an EU country sets up a private entity in the Cayman Islands and holds financial assets in Luxembourg through this entity. Once the private entity obtains the status of 'financial institution', the reporting duty is shifted from the financial institution in Luxembourg that maintains the financial accounts of the entity to the private entity itself. In this way, the third-party reporting turns into a self-reporting obligation where the taxpayers can opt for not reporting the overseas income to the tax authority, as evident in the 'shell bank' case. However, after the implementation of DAC6, the moment a client enters into or is advised on a transaction that features elements of tax aggressiveness, the intermediary has to report all the information about it to the tax authority. If a financial institution is providing a service that is used by the client in connection to a reportable cross-border arrangement (e.g. accepting wire transfers of large amounts into accounts owned by private entities falling outside the CRS due diligence requirements), the financial institution has to report to the local tax authority information on the client's identity and the details on which type of transaction has been used, and its value.³ In this way, every EU tax authority would obtain detailed information on any cross-border transaction that is used to circumvent the automatic information exchange agreements and to obscure true beneficial ownership (see Hallmark D of Council Directive 2018/822/EU).

Moreover, regulating the conduct of intermediaries by mandating the disclosure of their clients' tax schemes can have a deterrent effect. The EU commission stated that it 'should be expected that the mandatory disclosure of potentially aggressive tax planning schemes would dissuade intermediaries from designing and marketing such schemes' (European Commission 2017). Advising on tax-aggressive arrangements after DAC6 involves increased compliance and reputation costs, as well as a higher detection risk (Noked and Marcone 2022). Thus, DAC6 has the power to make the involvement in such tax schemes less attractive.

We focus our analysis on how the introduction of DAC6 affects cross-border deposits of EU residents, who are potentially experiencing a change in the detection risk, and compare their behaviour to the one of non-EU OECD residents who are unaffected by DAC6, but face a similar economic and fiscal environment. For residents in EU countries, we expect a reaction to the new disclosure requirements (Allingham and Sandmo 1972; Slemrod 2019). If DAC6 is able to crack down on offshore tax evasion, we should expect an increase in the incentives to report previously undisclosed offshore deposits in EU countries.

We address this question by investigating the direct effect of DAC6 on EU resident behaviour and the indirect effect of DAC6 on the use of citizenship- and residence-by-investment schemes. Specifically, we estimate tax evaders' reaction to a mandatory disclosure rule of aggressive tax arrangements at a within-country time and country-pair level by using a difference-in-difference (DiD) design. We follow the related literature on cross-border tax evasion (e.g., Casi et al. 2020; Huizinga and Nicodéme 2004; Johannesen and Zucman 2014; Menkhoff and Miethe 2019) and proxy tax evasion behaviour by considering the outstanding volume of cross-border deposits placed in tax havens. The data originates from the Bank for International Settlements (BIS). Our sample period is from the first quarter of 2017 to the last quarter of 2019 to avoid confounding events affecting cross-border deposit movements like the introduction of the CRS and the COVID-19 pandemic.⁴

We begin our analysis by investigating the direct effect of DAC6 on tax evaders by focusing on EU residents, who are the ones experiencing the change in detection risk. We compare their behaviour pre-

³ For more details on the implications of DAC6 for financial institutions, see https://www.ey.com/en_be/financial-services/areyou-ready-for-dac-6.

⁴ We offer evidence on a longer time period in Appendix A1.

and post-DAC6 to that of non-EU OECD residents as the latter is a suitable control group in our sample. Non-EU OECD residents (henceforth non-EU residents for convenience) face a similar economic and tax environment and are, at the same time, not affected by DAC6 because the information collected under DAC6 is not exchanged with their respective country of residence. We estimate the changes in cross-border deposits of EU residents compared to those of non-EU residents pre- and post-DAC6, combining a regression analysis and an event study design. Our results indicate that deposits of EU residents in the EU increase by 11 per cent post-DAC6, while no change in cross-border deposits outside the EU is detected, suggesting that the policy is effective. When considering the economic relevance of our main estimate, DAC6 led to an increase of US\$124 billion in cross-border deposits held by EU residents in the EU.

However, we also show that DAC6 is still far away from a perfect policy tool against tax evasion. We highlight this by investigating cross-country differences in regulatory environments. First, countries differ in monetary penalties. Usually countries impose a maximum monetary penalty of less than \in 100,000 per non-compliant institution, regardless of the value and volume of the non-disclosed or wrongly disclosed transaction (Casi et al. 2021). Only Spain refrains from setting an upper limit and instead charges a penalty that is proportional to the value of the incorrectly reported transaction or to the related intermediary fee. Our results show a statistically significant decrease in deposits of approximately 16 per cent in Spain post-DAC6, suggesting tax evaders withdraw funds from countries where compliance under DAC6 is expected to be higher given the high penalty.

Second, countries differ in the definition of who enjoys the legal professional privilege with respect to the reporting duty. Most EU member states restrict it to lawyers, tax advisers, and accountants, but a few extend it to financial institutions. Enjoying the legal professional privilege impacts the actual obligations under DAC6 since the reporting duty is shifted from the intermediary to the taxpayer.⁵ The detection risk under DAC6 is, therefore, arguably reduced in those countries where the reporting obligation for an intermediary is restricted, as evident from the 'shell bank' case (US Senate Finance Committee 2022). According to national law, France extends the legal professional privilege to financial intermediaries. Our findings suggest that the generous scope for the legal professional privilege granted by France induced a relocation of income and wealth. Specifically, we find an approximately 30 per cent increase in cross-border deposits in France post-DAC6.

In the second part of our study we analyse whether taxpayers are able to circumvent reporting under DAC6 altogether by exploiting citizenship-by-investment programmes (CBI) or residence-by-investment programmes (RBI). Anecdotal evidence suggests that such schemes have been used for tax evasion purposes, especially to circumvent the reporting duty under the automatic exchange of information (AEOI) agreements (e.g. under the CRS; Christians 2017; European Parliament 2016; Mehboob 2019). In the context of DAC6, an EU taxpayer could make use of multiple citizenship/residence rights to channel a cross-border arrangement outside the EU and, in this way, avoid the reporting duty under DAC6. This is possible because the access to multiple citizenship and residence rights enables tax evaders to select the country where the transaction originates. Our results show a statistically significant increase of 30 per cent in cross-border deposits owned by residents of CBI/RBI countries held outside the EU compared to residents of non-CBI/RBI countries post-DAC6. When considering the economic size of the effect we detect, this translates into an approximately US\$14 billion increase of cross-border deposits held by CBI/RBI residents in the deposit locations outside EU post-DAC6, of which US\$7 billion are held in tax havens outside the EU. Our findings provide evidence of the use of these schemes as regulatory arbitrage to circumvent the disclosure mandated under DAC6.

⁵ Anecdotal evidence shows that whether or not a client's adviser enjoys the legal professional privilege affects the final decision over which company to hire for tax planning services. Specifically, in EU member states where the legal professional privilege for DAC6 reporting duty is restricted to lawyers, the big four accounting firms have experienced a significant loss of clients, who have moved to law firms for tax advisory service. For more details, see Haines (2020).

We provide several robustness checks for corroborating the validity of our main results. We run a placebo test where we only consider residents of Ireland, Portugal, and the UK since these countries had a similar disclosure rule already in place before DAC6, and we confirm that, as expected, no statistically significant difference in cross-border deposits between treatment and control group occurs. In addition, we validate that our baseline results are not driven by the control group in a split test where results for the treatment and control groups are plotted separately. We also test the reliance of our identification on our choice of fixed effects, by modifying the fixed effect structure of our main results. Finally, we prove the validity of our results using a longer sample period.

Overall, our study contributes to the literature on regulations targeting illicit financial flows. So far, mixed results have emerged on the effectiveness of tax transparency in curbing tax evasion. While tax evaders reacted to the agreements to exchange information by reducing wealth and related income in cooperative jurisdictions, these funds haven't been repatriated but have been reallocated to tax havens not covered by such information exchange agreements (e.g., Casi et al. 2020; De Simone et al. 2020; Johannesen and Zucman 2014; Menkhoff and Miethe 2019). An MDR like the one under DAC6 is considered to have the potential to close all loopholes in the existing global tax transparency framework (Noked and Marcone 2022). Our study shows that DAC6 has certain important elements of weaknesses. Importantly, we document the use of CBI/RBI as a regulatory arbitrage strategy to escape the reporting of information on the use of cross-border transactions used for aggressive tax planning purposes.⁶

We also contribute to the literature on tax enforcement. Overall, government attention to tax compliance increased largely after the financial crises as a result of the substantial deficits (International Monetary Fund 2015: 6) and enhancing tax compliance is now a top priority for policy-makers, given the massive economic shocks from the COVID-19 pandemic. There is a vast literature on studying the effect of stricter enforcement rules on tax compliance (e.g., Almunia and Lopez-Rodriguez 2018; Carrillo et al. 2017; Fack and Landais 2010; Kleven et al. 2011; Kopczuk et al. 2016). We focus on cross-border tax evasion and investigate the effect of mandating the automatic collection and exchange of information on aggressive tax arrangements on under-reported income and wealth held abroad. Specifically, we address the call from Slemrod (2019) on understanding the role of tax professionals in administration and enforcement, as well as the importance of the penalty level for non-compliance.

Finally, the results of our study inform policy-makers, given the current global debate on the necessity to revise the rules for tax advisory services.⁷ Our study sheds light on the effectiveness of an increasing disclosure mandate for intermediaries on their clients' tax schemes. We provide evidence of the relevance of imposing sufficiently high enforcement to ensure compliance and of restricting the professional legal privilege to ensure third-party reporting. In doing so, we offer important insights to the EU member states as well as to those countries outside the EU that have similar mandatory intermediary disclosure requirements or are considering introducing them. We also contribute to the international debate on the risk related to CBI/RBI programmes. Both the EU and the OECD have expressed concerns on the misuse of these programmes (European Parliament 2018; OECD 2020). The results of our study offer novel empirical evidence to support the policy debate in this area.

The rest of the paper is organized as follows. Section 2 outlines the institutional background. Section 3 develops our testable hypotheses. Section 4 describes our data and research methodology, while Section 5 presents our main empirical results. Section 6 summarizes the robustness checks and Section 7 concludes.

⁶ Langenmayr and Zyska (2021) provide evidence of increased volume of illicit financial flows in connection to the introduction of CBI/RBI programmes.

⁷ Recently the EU Commission launched a consultation on the topic; see https://www.etaf.tax/index.php/newsarea/300-weekly-tax-news-11-july-2022.

2 Institutional background

2.1 The EU's mandatory disclosure rule

In 2010, the introduction of FATCA in the United States enabled the development of an extremely powerful standard for the AEOI in tax matters. This policy tool was enacted with the ambition to overcome the weakness of previous initiatives in the field and, in this way, to finally put an end to the substantial tax revenue loss resulting from US citizens hiding income and wealth offshore. It obliges foreign financial institutions to collect financial account information on behalf of their clients if they are US citizens and to automatically transmit them to the IRS. The introduction of FATCA pushed an international discussion at the OECD level on developing a global standard for the AEOI. On 21 July 2014, the OECD published the final version of the global standard for automatic exchange of financial account information in tax matters: the CRS. Currently, more than 100 jurisdictions around the world have implemented the CRS. Given the broad scope and the extensive country coverage, the AEOI information system under FATCA and the CRS presents certain key features that make it substantially different from any initiative launched in the field so far. The true revolution in the level of scrutiny of illicit financial flows held overseas would considerably redesign the cross-border tax evasion schemes detected so far. And, indeed, FATCA and the CRS have been successful in reducing international tax evasion via tax havens, resulting in significant additional reporting of accounts (e.g., Casi et al. 2020; De Simone et al. 2020; Menkhoff and Miethe 2019), especially of the highest-earning individuals (Johannesen et al. 2020). However, indirect evidence also suggests that wealth and income are relocated to non-reportable assets such as real estate and luxury goods, and to non-cooperative jurisdictions (e.g., Bomare and Herry 2022; Casi et al. 2020; De Simone et al. 2020).

In June 2018, the sixth amendment to the Directive on Administrative Cooperation was introduced with the ambition to close the loopholes detected in previously launched tax transparency initiatives. The Directive on Administrative Cooperation (Council Directive 2011/16/EU, or DAC1) is a legal instrument introduced at the EU level in 2011 with the aim to increase the automatic collection and exchange of information across EU tax authorities. Under DAC1, information on every type of tax, other than VAT, customs duties, excise duties, and social security contributions could be exchanged upon request across member states within a six-month period. In this way, DAC1 ensures that the OECD standard for the exchange of information on request is implemented in the EU. Subsequent amendments to DAC1 introduced global tax transparency initiatives within the EU. Specifically, in 2014, Council Directive 2014/107/EU (DAC2) introduced the CRS at the EU level. In 2015, Council Directive 2015/2376/EU (DAC3) introduced a proper definition of advanced cross-border rulings as well as advanced pricing arrangements, and imposed the AEOI of those. In 2016, Council Directive 2016/881/EU (DAC4) introduced the requirement for country-by-country reports at the EU level as proposed under Action 13 of the Base Erosion and Profit Shifting (BEPS) project.⁸ Council Directive 2016/2258/EU (DAC5), introduced in 2016, forces local financial institutions to identify and report to the respective tax authority the information on the beneficial owner of an intermediary structure.

Revelations from tax scandals such as the Paradise Papers and the Panama Papers publicized by the International Consortium of Investigative Journalists raised concerns regarding the pervasive use of harmful tax practices and the necessity to strengthen the fight against tax evasion and tax avoidance (European

⁸ Empirical evidence on the impact of DAC2 and DAC3 provide mixed evidence on their effectiveness. Casi et al. (2020) show that the introduction of the CRS at the EU level did not trigger a strong reaction from tax evaders given the existence of the automatic collection and exchange of interest income under the Savings Directive. Instead, the duty of large corporations to disclose country-level economic activity to tax authorities, such as the one mandated under DAC3, induced a reallocation of investment to the EU, but mainly in those countries offering preferential tax regimes (Olbert and De Simone 2021). Overall, only weak evidence of improved tax compliance has been detected (Joshi 2020).

Parliament 2016). In particular, the role certain financial institutions and other intermediaries played in supporting clients to establish complex legal structures, with the only intent being to evade tax obligations, emerged. At the informal meeting of the Economic and Financial Affairs Council in April 2016, member states welcomed initiatives, such as those stated in BEPS Action 12, requiring taxpayers and advisers to disclose aggressive tax planning arrangements.⁹ As a result, the EU adopted Council Directive 2018/822/EU (DAC6) in May 2018, which represents the sixth amendment to the Directive on Administrative Cooperation.

Under DAC6, a comprehensive set of information on the cross-border arrangements needs to be reported to the local tax authorities if certain criteria are met.¹⁰ One of the most important criteria is that the crossborder arrangement needs to involve at least one EU country. Moreover, cross-border arrangements have to be reported if they display certain predefined characteristics, called 'hallmarks'. The definition of the selected hallmarks mainly reflects those in Action 12 of the BEPS initiative and includes generic and specific hallmarks. In particular, generic hallmarks include all arrangements embracing three elements: confidentiality, intermediary fee, and standardized documentation. The generic hallmark must be considered only if it can be proven that the main benefit of a cross-border arrangement is to gain a tax advantage. If this occurs, the main benefit test is satisfied, and the cross-border arrangement must be reported to the respective authority. Specific arrangements include four types of arrangements. First, it embraces all arrangements that satisfy the main benefit test and enables the taxpayer to use losses to reduce a tax liability, to convert income into capital, gifts, or other categories of revenue, which are taxed preferentially, or to exploit circular transactions resulting in round-tripping of funds. Second, it comprises all cross-border arrangements that aim at circumventing the requirements under AEOI legislation or agreements across EU member states. Third, it encompasses all cross-border arrangements involving deductible cross-border payments between two or more parties if certain conditions are met, namely depreciation of the same asset in multiple jurisdictions, multiple relief from double taxation on the same claims for more than one taxpayer, or transfers of assets where significant discrepancies exist in the amount being treated as payable with respect to the assets in those jurisdictions involved. Lastly, it concerns transfer pricing and covers all arrangements that do not comply with the arm's-length principle or the OECD transfer pricing guidelines, or are within the scope of the AEOI on advance cross-border rulings but are neither reported nor exchanged.

A comprehensive set of information on the reportable cross-border arrangements is due within 30 days from the date the scheme is made available. Information to be reported includes a summary of the content of the arrangement, the value of the arrangement, the category of hallmark to which it belongs, the identification detail of the intermediaries, and the relevant taxpayers. In this way, the type of information disclosed under DAC6 is substantially different from that obtained under the CRS and FATCA, as illustrated in Table 1.¹¹

Under the CRS, tax authorities obtain extensive information on financial assets held in foreign jurisdictions that are participating in the AEOI. Yet, anecdotal and empirical evidence (e.g., Bomare and Herry 2022; Casi et al. 2020; De Simone et al. 2020) suggests that wealth and income have been relocated to non-reportable jurisdictions and non-reportable assets. DAC6 has been introduced with the aim to close such loopholes and ensure that no possibility of escaping the AEOI within the EU is exploited. This is achieved by mandating the reporting of specific types of transactions that are considered to facilitate the

⁹ For more details on BEPS Action 12, see (OECD 2013: 22–23).

¹⁰ For a comprehensive overview of DAC6 and its national implementation across EU member states, see Casi et al. (2021).

¹¹ In Table 1 we focus our comparison on how DAC6 differs from the CRS. For this reason, we exclusively explain the type of transactions that need to be reported under Hallmark D of DAC6—that is, the category of transactions that potentially undermine the AEOI under the CRS.

circumvention of the reporting duty under the CRS.¹² The moment a client is advised on a type of transaction listed under DAC6, the adviser has to report all the information about it, including the information on the client receiving the advice and the details on type of transactions, and their value.¹³

Panel A	Common Reporting Standard (DAC2)
The scope	Information on financial assets held outside the country of residence
Who gets it	Information is exchanged with all CRS-participating jurisdictions
What is obtained	Automatically exchanged information:
	 Identification information of the account holder; if indirectly owned, on the last beneficial owner
	• Financial information on the account, including the balance, the interest and/or dividend amount,
	the amount of other income generated with respect to the assets held in the account, the proceeds
	from the sale or redemption of financial assets, the amount paid or credited by the reporting financial
	institution in reference to the account
Panel B	Mandatory Disclosure Rule (DAC6)
The scope	Hallmark D: arrangement which may have the effect of undermining the reporting obligation under
	the CRS, including transfer of funds to non-participating jurisdictions, non-reportable assets, and
	non-reportable financial institutions
Who gets it	Information is exchanged with EU member states only
What is obtained	Automatically exchanged information:
	• Identification information of the intermediaries, and of the relevant taxpayers involved in the re-
	ported arrangement
	• A summary of the content of the arrangement, the value of the arrangement, the category of hall-
	mark to which it belongs, the identification detail of the intermediaries, and of the relevant taxpayers

Table 1: Comparison of the CRS (DAC2) and MDR (DAC6)

Source: authors' compilation.

Besides the establishment of shell entities as in the 'shell bank' case, another reportable transaction includes the transfer of income and wealth to a jurisdiction that is not subject to the CRS. Imagine, for instance, a German taxpayer that is transferring its bank account from the Cayman Islands to the Dominican Republic after the CRS is introduced to avoid that German tax authorities automatically obtain information on it, as the Cayman Islands participates in the CRS while the Dominican Republic does not. After the introduction of DAC6, the German tax authorities would obtain extensive information on the tax advisers and to which clients this type of transaction is advised. Likewise, transactions involving the transfer of funds to 'non-reportable financial assets' and/or 'non-reportable financial institutions', as an attempt to avoid the CRS reporting duty, have to be reported to the respective tax authority. Examples that qualify as non-reportable financial assets and non-reportable financial institutions are the use of certain types of virtual currency or derivatives contracts and the use of trusts, which under certain conditions automatically qualify as active non-financial entities. Thus, DAC6 probably makes transactions that aim to circumvent the AEOI no longer attractive and allows EU tax authorities to close existing loopholes.

2.2 DAC6 and citizenship/residence-by-investment

CBI/RBI programmes have been introduced to attract funds from wealthy investors. Such programmes offer individuals the possibility to obtain citizenship or residence rights through local investment or against a flat fee. Survey evidence suggests that among various reasons, individuals opt for entering a CBI/RBI programme for tax planning reasons.¹⁴ Anecdotal evidence has highlighted that identity cards and similar documentation obtained under CBI/RBI programmes have been misused to escape

¹² For the complete list, see page 12 of Council Directive 2018/822/EU.

¹³ For more explanation, see EY (2020).

¹⁴ Specifically, 3 per cent of the surveyed individuals who are interested in CBI/RBI programmes answered that the reason is tax-related. Other reasons include education access for children, better lifestyle, and higher security. For more details, see (OECD 2018: 58).

the reporting duty under the CRS.¹⁵ Thus, one could expect that the availability of multiple citizenship and/or multiple residence rights could represent a channel to escape the reporting duty under DAC6. The simplified example shown in Figure 1 clarifies the mechanism.



Figure 1: Circumventing the DAC6 reporting requirement

Note: this figure provides an example of escaping DAC6 reporting duty using the United Arab Emirates' CBI programme. Source: authors' compilation.

Assume, for example, an EU taxpayer that set up an investment entity in the Dominican Republic through a local bank account after the introduction of CRS. After the implementation of DAC6, every EU tax authority would become aware of every transaction the EU taxpayer conducted involving an EU member state and a CRS non-participating country, such as the Dominican Republic. Yet, the tax evader could enter into a transaction in the Dominican Republic using the passport or residence certificate from a CBI/RBI country such as the United Arab Emirates and, in this way, circumvent the duty to report the transaction under DAC6 to the true country of residence.

3 Hypotheses development

The overall objective of the MDR under DAC6 is to guarantee that tax authorities receive early information on cross-border arrangements, which could potentially pose a risk of being aggressive from a tax perspective (European Council 2017). Tax authorities within the EU obtain immediate exhaustive information on all potentially tax-aggressive transactions at the time they are ready to be used by the taxpayers or to be promoted by the intermediary. Post-DAC6, EU tax authorities would obtain information on transactions involving opening bank accounts in countries that do not exchange information under the CRS. Because DAC6 enhances detection risk of holding income and wealth in CRS non-participating countries, we expect a reduction of cross-border deposits in such countries. Since DAC6 prevents individuals escaping into non-reportable financial assets and/or non-reportable financial institutions, we expect an increase in cross-border deposits in EU deposit locations. We summarize in the following hypothesis:

Hypothesis 1 The introduction of DAC6 leads to an increase [decrease] of cross-border deposits in EU countries [CRS non-participating countries] by EU residents.

Yet, we expect that DAC6 does not equally impact EU taxpayers given certain key differences detected by Casi et al. (2021) when analysing its local implementation. First, monetary penalties for misreporting vary substantially across EU member states. Specifically, the enforcement level has been detected to be low across EU member states with the exception of certain countries, including Spain and Poland. Spain is the only country that opted for strong enforcement as the penalty is based on the value of

¹⁵ See, for example, Christians (2017), Mehboob (2019), European Parliament (2018), and OECD (2020). See also Langenmayr and Zyska (2021) for an analysis that shows how CBI schemes can be an effective tool for tax evaders to avoid the increased detection risk under AEOI.

the incorrectly reported or non-reported transaction or to the related intermediary fee. While Poland charges up to \in 5 million for misreporting by an individual under DAC6, we do not have the deposit data in Poland and therefore only explore the effect in Spain. Hence, we expect that tax evaders decrease the level of income and wealth only in those countries where enforcement under DAC6 is strongly established.

Second, under DAC6, the primary duty to report is on intermediaries. However, even if the cross-border arrangement occurs within the EU territory, the information on a reportable cross-border arrangement needs to be transmitted by the EU-based taxpayer if (1) the intermediary is not located in the EU, (2) the intermediary is restricted by professional privilege or secrecy rules, or (3) the arrangements occurred inhouse. France, Austria, and Malta extend the legal professional privilege to financial institutions, which means that in these countries the taxpayer is responsible for reporting under DAC6. We, therefore, expect that tax evaders increase the level of income and wealth in those countries with a broad scope for the legal professional privilege. Overall, we summarize our discussion in the following hypothesis:

Hypothesis 2 The introduction of DAC6 decreases [increases] cross-border deposits of EU residents in EU countries with a strict [more lenient] regulatory environment.

While cross-country differences in regulatory environments provide individuals with an avoidance opportunity through the reallocation of wealth to favourable environments permissible with the DAC6 reporting requirements, CBI/RBI programmes enable individuals to continue the evasion of taxes by escaping reporting under DAC6 altogether. Such programmes offer individuals the possibility to obtain citizenship or residence rights through local investment or against a flat fee, which allows those individuals to invest through countries offering CBI/RBI programmes. Thus, the availability of multiple citizenship and/or multiple residence rights represents a channel to escape the reporting duty under DAC6 by disguising an individual's true residence. We expect cross-border deposits from CBI/RBI residents to increase only outside the EU because DAC6 mandates the reporting of cross-border transactions with an EU nexus (i.e. also those conducted by non-EU residents in the EU). While tax havens are naturally a place where we would expect cross-border deposits from RBI/CBI programmes to flow, recent evidence suggests that individuals hide their wealth also in non-tax haven locations. For example, several luxury properties—including yachts and houses—belonging to sanctioned Russian oligarchs were located in high-tax countries such as Norway or the United States (Harding 2022; Lambert 2022). We summarize in the following hypothesis:

Hypothesis 3 The introduction of DAC6 increases cross-border deposits in non-EU countries that originate from CBI/RBI countries.

4 Research design

4.1 Data

The data on the cross-border deposits are obtained from the Bank of International Settlements-Locational Banking Statistics (BIS-LBS) database (BIS 2020). The BIS offers bilateral quarterly data on deposits held by individuals and entities that are not residents of the country where the reporting bank is located. From the BIS-LBS data we observe cross-border deposits held by the residents of 215 countries (resident countries) in a select list of 31 countries (deposit locations). We retain all country-pairs for the purpose of our main analysis, but we exclude cross-border inter-bank deposits because they are not identified as a channel for tax evasion (Johannesen and Zucman 2014). Although the data exhibit certain limitations

in terms of coverage and granularity,¹⁶ they are extensively used in the literature on cross-border tax evasion because they offer a sound proxy for capturing the reaction of tax evaders to increased scrutiny (e.g., Casi et al. 2020; Johannesen and Zucman 2014; Langenmayr and Zyska 2021; Miethe 2020)). Specifically, the BIS-LBS data enable us to observe, for example, the total amount of deposits French residents own in active banks located in 31 deposit countries, including several well-known tax havens. Moreover, we limit the period of analysis to the first quarter of 2017 until the fourth quarter of 2019. This allows us to exclude possible confounding impacts of the introduction of the CRS and the global pandemic.

For our analysis of the effect of the DAC6 on cross-border deposits owned by EU residents, we limit our sample to residents of all EU and OECD member states to ensure high cross-country comparability across the treatment and the control group. We exclude residents of Malta and Cyprus because these are the only EU member states offering highly risky CBI/RBI programmes as defined by the OECD (2018), and such programmes represent a possible channel to circumvent the reporting duty under DAC6.

Moreover, to ensure that our results are not driven by the reactions of multinational companies, we gather BIS-LBS data on the sectoral decomposition of cross-border deposits.¹⁷ More specifically, we observe the volume of deposits owned by banks, non-bank financial institutions, non-financial corporations, households, and general government at the aggregated level. This means that we only observe the total volume of corporate-owned deposits in a country, but we don't know the country of residence of the corporation that owns the deposit. For this reason, we cannot directly take advantage of the data, but instead use the sectoral decomposition to infer the composition of cross-border deposits in the deposit country considered in our analysis. The sectoral decomposition is only available for a limited number of deposit locations and we keep those for which deposits owned by non-financial corporations is less than 50 per cent of total non-bank deposits.¹⁸ Our deposit locations include Austria, Australia, Canada, Denmark, France, Italy, Spain, Sweden, Switzerland, South Africa, the UK, and the United States.

We observe that the data from BIS-LBS has missing values for the deposits in certain quarters for some country-pairs. To balance the panel data, we omit all country-pair observations with missing values for deposits in any given time period. Table 2 presents the summary statistics for cross-border deposits owned by EU versus OECD residents. During our sample period (2017Q1–2019Q4), the average bilateral cross-border deposit volume amounts to US\$6.6 billion. The value of cross-border deposits owned by EU residents (the treatment group) is comparable to that owned by non-EU OECD residents (the control group) in terms of average bilateral cross-border deposit with EU countries and non-EU countries. The descriptive statistics support our assumption that non-EU OECD residents are a sound control group for our analysis.

Figure 2 shows the development of cross-border deposits over time owned by EU residents and non-EU residents. Overall, the volume of cross-border deposits owned by EU residents is higher than that owned by non-EU residents. At the aggregated level there is no visible change in the level of cross-border deposits post-DAC6 from either EU or non-EU residents. In the empirical analysis that follows we will provide country-level evidence of the reaction of tax evaders to the new mandatory disclosure rule.

¹⁶ See Casi et al. (2020) for an overview of the data limitations.

¹⁷ Empirical evidence from Edwards et al. (2021) shows that DAC6 increases the effective tax rates of EU multinationals or multinationals with EU subsidiaries, suggesting that the new MDR has been successful in reducing tax avoidance.

¹⁸ We run robustness checks where we keep all deposit locations and alternatively where we only keep deposit locations where one-third or less of the deposits are owned by non-financial corporations. These changes do not affect our inferences, which suggests that multinationals' reaction to DAC6 does not drive our results.

Table 2: Summary statistics

Deposit in	All residents		EU residents		Non-EU residents		
	Obs. Mean SD		Obs.	Mean	Obs.	Mean	
Cross-border deposits	5,436	7,063	40,027	3,396	6,531	2,040	8,248
Of which: In the EU	3.276	8.064	45.415	2.004	6.730	1.272	10.103
Outside the EU	2,160	5,544	30,013	1,392	5,747	768	5,175

Note: the table presents summary sample statistics on bilateral cross-border deposits in millions of US dollars. Data from 2017Q1 to 2019Q4. Deposits are from EU and OECD residents in all reporting countries. Deposits are further split based on the deposit location (in the EU and outside the EU).

Source: authors' compilation based on data from the BIS-LBS, Table A6.

Figure 2: Time trends: EU- and OECD-owned deposits



Note: the graph shows the evolution of cross-border deposits owned by EU residents (black line) and non-EU OECD residents (blue line). The red dashed line represents the introduction of DAC6. Source: authors' compilation based BIS-LBS, Table A6.

For our analysis of the effect of DAC6 on cross-border deposits owned by CBI/RBI countries, we collect information on CBI/RBI programmes from the OECD website.¹⁹ The OECD classify those programmes as being high risk if they offer access to a low income tax rate on financial assets and do not impose any physical presence in the country for a significant amount of time. The list of countries we consider is United Arab Emirates, Bahamas, Bahrain, Barbados, Cyprus, Dominica, Grenada, Malta, Saint Lucia, Turks and Caicos Islands, and Vanuatu.

Our sample extends to all bilateral data available at the BIS. We only exclude residents of EU member states from our control group (we include non-OECD countries as well) as they are directly affected by the DAC6. We still retain residents of Malta and Cyprus, since those two countries offer high-risk CBI/RBI schemes.²⁰ Moreover, we retain all deposit locations because there is no reason to expect that any detected movement of cross-border deposits in this part of the analysis is driven by multinational

¹⁹ For more information, see https://www.oecd.org/tax/automatic-exchange/crs-implementation-and-assistance/residencecitizenship-by-investment.

²⁰ We test the effect of DAC6 separately for EU and non-EU CBI/RBI residents and find similar results.

companies reacting to DAC6. It is reasonable to expect that any increase in cross-border deposits held by residents of aggressive CBI/RBI countries is exclusively driven by individual incentives to hide the true citizenship/residency to avoid the higher detection probability post-DAC6.

Table 3 presents the summary statistics for cross-border deposits owned by CBI/RBI versus non-CBI/RBI residents. During our sample period (2017Q1–2019Q4), the average bilateral cross-border deposit volume amounts to US\$0.5 billion. The value of cross-border deposits owned by CBI/RBI residents (the treatment group) is significantly lower compared to the one owned by non-CBI/RBI residents (the control group), but have a similar pre-DAC6 development as visible from Figure 3. Yet, cross-border deposits owned by CBI/RBI residents experience an increasing trend post-DAC6, while the ones owned by non-CBI/RBI countries do not.

Deposit in	A	All residents		CBI/RBI residents		Non-CBI/RBI residents	
	Obs.	Mean	SD	Obs.	Mean	Obs.	Mean
Cross-border deposits	39,096	1,080	15,237	2,508	517	36,588	1,119
Of which: In the EU Outside the EU	18,756 20,340	1,130 1,034	41,630 12,386	1,176 1,332	482 548	17,580 19,008	1,174 1,068

Note: the table presents summary sample statistics on bilateral cross-border deposits in millions of US dollars. Data from 2017Q1–2019Q4. Deposits are from non-EU residents as well as from residents of CBI/RBI countries in all reporting countries. The deposits of residents of Malta and Cyprus (both are CBI/RBI as well as EU countries) are also included when calculating the statistics. Deposits are further split based on the deposit location (in the EU and outside the EU).

Source: authors' compilation based on data from the BIS-LBS, table A6.

Figure 3: Time Trends: CBI- and RBI-owned deposits



Note: the graph shows the evolution of cross-border deposits owned by CBI/RBI residents (black line) and non-CBI/RBI residents (blue line). The red dashed line represents the introduction of DAC6. The non-CBI/RBI countries include all the countries except EU countries.

Source: authors' compilation based on data from the BIS-LBS, table A6.

4.2 Methodology

Our analysis is based on both a DiD estimation and an event study approach. We use the DiD design to estimate the average effect of the DAC6 on cross-border deposits held by EU residents. We run

regressions of the form:

$$Deposits_{ijt} = \alpha + \beta_2 PostDAC6_{jt} \times EUResidents_i + \gamma_{jt} + \theta_{ij} + \varepsilon_{ijt}$$
(1)

We use the event studies to explore pre-trends and dynamic effects of the DAC6 on cross-border deposits held by residents of EU versus non-EU countries, which formally reads as:

$$Deposits_{ijt} = \sum_{k=-4}^{4} \alpha_k D_{jt}^k \times EUResidents_i + \gamma_{jt} + \theta_{ij} + \varepsilon_{ijt}$$
(2)

In both specifications, the dependent variable *Deposits_{ijt}* is the natural logarithm of the volume of cross-border deposits located in country *j* and owned by a resident of country *i* at the end of quarter *t*. *EUResidents_i* is a dummy taking value 1 when the resident country is an EU member state. The variable of interest in the DiD specification is the interaction of *PostDAC6_{jt}* and *EUResidents_i*. *PostDAC6_{jt}* is the post-period dummy and it switches to 1 after DAC6 became effective in June 2018.²¹ The variables of interest in the event study are the dummies D_{jt}^k , indicating a point in time *k* periods from the DAC6 treatment and interacted with *EUResidents_i*. As is the standard in the literature for event studies, we omit the indicator for period t - 1, which serves as a benchmark. We bin the treatment indicators at the endpoints.²² We include deposit–country quarter–year fixed effects γ_{it} to control for common time trends affecting cross-border deposits (e.g., globalization of financial markets and economic shocks) and deposit country-specific demand-side shocks. Ordered country-pair fixed effects θ_{ij} are added to control for all time-invariant country-pair factors (e.g., distance, common language), which might affect the change in cross-border deposits as a reaction to DAC6. Our standard errors are cluster-robust, with clustering at the resident country level. The error term is denoted by ε_{iit} .

Similarly, we use a DiD regression followed by an event study design to estimate the effect of the DAC6 on cross-border deposits owned by CBI/RBI residents in the EU and outside the EU. Specifically, we compare the changes in cross-border deposits held by residents of CBI/RBI countries to the ones held by residents of non-CBI/RBI countries (excluding EU countries) pre- and post-DAC6 implementation. The regression equations are the same as equations (1) and (2), but we substitute the treatment variable with $CBIRBI_Residents_i$, which takes a value of 1 when the resident country *i* is a CBI/RBI country.

5 Results

5.1 The effect of DAC6 on EU-owned deposits: a cross-country analysis

Main analysis

In this section we show the results of testing hypotheses 1 and 2. Table 4 illustrates the results for the DiD regression model from equation (1). Columns (1)–(3) refer to the results of testing hypothesis 1, whereas Columns (4)–(6) report the results of testing hypothesis 2. Column (7) serves as a contrast to Columns (4)–(6) by showing the results for the countries not offering a preferential or stricter regulatory environment.

²¹ EU member states had until 31 December 2019 to transpose the directive into national law. Yet, every national law had a retrospective element since all cross-border arrangements advised or in use from 25 June 2018 needed to be reported if fulfilling the characteristics stated under DAC6.

²² Binning implies here that the indicator t - 4 stands for treatment at time t - 4 or more periods ago and the indicator t + 4 stands for time t + 4 or more periods in the future. See Schmidheiny and Siegloch (2019) and Fuest et al. (2018).

	Table 4:	The effect	of DAC6	on EU-owned	deposits
--	----------	------------	---------	-------------	----------

Cross-border deposits							
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Sample	Within	Outside	United	Spain	France	France	Other
	EU	EU	States			and Austria	EU
Post-DAC6 \times EU residents	0.105**	-0.021	-0.058	-0.179*	0.265**	0.248***	0.105*
	(0.049)	(0.070)	(0.075)	(0.097)	(0.102)	(0.083)	(0.060)
Constant	5.378***	5.185***	7.075***	5.145***	6.713***	5.674***	5.281***
	(0.017)	(0.026)	(0.029)	(0.035)	(0.036)	(0.029)	(0.021)
Observations	3,276	2,160	480	492	492	984	1,800
R-squared	0.990	0.988	0.991	0.989	0.983	0.986	0.992
Country-pair FE	YES	YES	YES	YES	YES	YES	YES
Deposit-quarter-year	YES	YES	NO	NO	NO	YES	YES
Quarter-year	NO	NO	YES	YES	YES	NO	NO
Clustering Resident country							

Note: the table reports the main DiD estimates. The dependent variable is the log of cross-border deposits held by residents of country *i* in banks in deposit country *j* at the end of quarter *q*. The sample period is 2017Q1–2019Q4 and the sample is restricted to residents of EU and OECD countries. The EU residents indicator takes a value of 1 if the deposits are owned by a resident of an EU country and 0 otherwise. In Column (1), deposit locations consists exclusively of EU countries, while in Column (2), deposit locations are exclusively non-EU countries. In Column (3), deposit location is the United States, while in Columns (4), (5), (6), and (7) the deposit locations are Spain, France, France and Austria, and Denmark, the UK, Italy, and Sweden, respectively. In Columns (1), (2), (6), and (7), ordered country-pair and deposit country × quarter–year fixed effects are included. In Columns (3), (4), and (5) ordered country-pair and quarter–year fixed effects are included. Standard errors are clustered at the residence country level and are reported in parentheses. *** *p* < 0.01, ** *p* < 0.05, * *p* < 0.1. Source: authors' compilation based on BIS-LBS data.

In Column (1) the coefficient on the interaction term is positive and statistically significant, suggesting the introduction of DAC6 led to a 10 per cent increase in deposits in the EU by EU residents, most likely due to greater disclosure of offshore accounts. In a given quarter-year, the average amount of deposits held by EU residents in our sample of EU countries is US\$1,131 billion. Based on our estimates, this amount increased by US\$124 billion post-DAC6. Column (2) shows the results for non-EU deposit locations, indicating DAC6 did not affect deposits of EU residents located outside the EU as the coefficient on the interaction term is economically small and statistically insignificant. In Column (3) we focus exclusively on how DAC6 affects EU residents' US deposits. The reason is that the United States. is not participating in the CRS, which will make it easier for individuals to evade taxes as information exchange does not happen automatically. If deposits in the United States are directly owned by tax evaders, we should observe a decrease in these deposits after the introduction of DAC6. Yet, our results suggest that US deposits of EU residents are not affected by DAC6. One explanation for this result is that tax evaders do not directly own their US deposits, but indirectly through shell companies set up in tax havens. While we do not have sufficient data on tax haven deposits to directly test this channel, we indirectly shed light on this strategy by studying the effect of DAC6 on the use of CBI/RBI programmes in the next section.

Figure 4 shows the graphical results from the event study design of equation (2). In both panels of Figure 4, the vertical line between t = -1 and t = 0 indicates the implementation of the directive by the EU. The reference to DAC6 with the vertical line between t = -1 and t = 0 applies to all other event studies that follow as well. In Panel (a), we witness a constant upward trajectory of deposits in EU deposit locations post-treatment, surging in the period t = 2 and persisting in the remaining periods of the study. Panel (b), depicting the results of non-EU deposit locations, does not illustrate any significant movement of deposits post-DAC6.





(a) Deposits within the EU

(b) Deposits outside the EU

Note: the figure shows the coefficients, each of which marks the change in cross-border deposits held within the EU (a) and outside the EU (b) around the DAC6 event date (in event time). The sample is restricted to residents of EU and OECD countries. The plotted coefficients are those of the interactions of the EU resident indicator and eight separate indicator variables, each marking one quarter over the sample period relative to the quarter before the DAC6 treatment event date (t = 0). We bin the treatment indicators before t - 4 and after t + 4 at the endpoints and omit the indicator for period t - 1, which therefore serves as a benchmark and has a coefficient value of 0 (and no confidence interval). The figure plots the coefficient estimates together with their 95 per cent confidence intervals for the DAC6 event date. The dependent variable is the log of cross-border deposits. Ordered country-pair and deposit country \times quarter–year fixed effects are included in both panels. Standard errors are clustered at the residence country level.

Source: authors' compilation based on BIS-LBS data.

In Columns (4)–(6) of Table 4 we present the results for testing hypothesis 2 to verify whether the regulatory environment plays a role for deposit relocation to the EU after the introduction of DAC6. Specifically, in Column (4) we analyse how deposits located in Spain are affected by the introduction of DAC6, while Column (5) shows the same analysis for France. Spain has a comparably stricter enforcement of DAC6 as it imposes a high monetary penalty on non-compliance, whereas France has a comparably weaker enforcement of DAC6 as it offers a legal professional privilege to financial institutions. If stronger enforcement disincentivizes misreporting or non-disclosure, we should observe a lower inflow of deposits into Spain, but a higher inflow of deposits into France. Interestingly, the coefficient on the interaction term in Column (4) is negative (–16 per cent) and significant, which suggests an outflow of deposits from Spain. In contrast, we observe a positive (30 per cent) and significant effect in Column (5), implying a strong inflow of deposits to France.

In Column (6) we present the combined results for deposits in France and Austria. We perform this additional test to show the effect of DAC6 on legal professional privilege by including a country where the legal professional privilege also extends to financial institutions, although it is granted only under certain conditions. Thus, in Austria, the extent of the privilege is limited in comparison to France. Austria also represents a special case given the long history of bank secrecy, which resulted in the country being considered a tax haven. However, we note that the inclusion of Austria does not change the result and indicates that any form of professional privilege may be used to relocate deposits.

Column (7) presents the results for all other EU locations excluding Spain, France, and Austria. We observe a coefficient that is economically of the same magnitude (10.5 per cent) as compared to the benchmark in Column (1). The combination of results suggests that some deposits owned by EU residents in Spain are relocated after DAC6 within the EU, and potentially to France or Austria. Hence, these results lend credence to the fact that deposits are not repatriated to the residence country, but relocated to a country with a more favourable regulatory environment.



Figure 5: Dynamic effect of DAC 6 on EU-owned deposits: cross-country analysis



Note: the figure shows the coefficients, each of which marks the change in cross-border deposits held in (a) the United States, (b) Spain (c) France, (d) France and Austria, and (e) Denmark, the UK, Italy, and Sweden around the DAC6 event date (in event time). The sample is restricted to residents of EU and OECD countries. The plotted coefficients are those of the interactions of the EU resident indicator and eight separate indicator variables, each marking one quarter over the sample period relative to the quarter before the DAC6 treatment event date (t = 0). We bin the treatment indicators before t - 4 and after t + 4 at the endpoints and omit the indicator for period t - 1, which therefore serves as a benchmark and has a coefficient value of 0 (and no confidence interval). The figure plots the coefficient estimates together with their 95 per cent confidence intervals for the DAC6 event date. The dependent variable is the log of cross-border deposits. Ordered country-pair and quarter–year fixed effects are included in Panels (a)–(c). Ordered country-pair and deposit country quarter–year fixed effects are included in Panels (d) and (e). Standard errors are clustered at residence country level. Source: authors' compilation based on BIS-LBS data.

Figure 5 shows the event study results for the sub-samples presented in the last five columns of Table 4. Figure 5(a) shows for the sub-sample of the United States that the effect size is small in magnitude and statistically insignificant in any post-treatment period. Panels (b) and (c) show the event study results for the sub-samples of Spain and France, respectively. In both cases, the effect becomes significant with some delay, but remains significant at the end of the post-treatment period. Panel (d) presents the result

for the sub-sample of France and Austria, which is very similar to the one in Panel (c). Finally, Panel (e) illustrates the results for other EU countries with no significant change in the post-treatment period. Though we observe a positive trend in the post-treatment period, it is not significant at the 95 per cent significance level.

Placebo test

We conduct a placebo test exploiting the introduction of an MDR in three countries prior to DAC6. More specifically, Portugal introduced its MDR in 2008, the UK followed two years later in 2010, and Ireland another year later in 2011.²³ Thus, for the residents of those countries we do not expect any effect of DAC6. We test our hypothesis by comparing cross-border deposits of residents of Ireland, Portugal, and the UK to those of residents of non-EU countries before and after the implementation of DAC6.

Figure 6 shows the results of the event study design of equation (2) by changing the treatment group that now only includes the deposits of residents from Ireland, Portugal, and the UK. We verify the effect of DAC6 on deposits in four different samples of deposit locations, namely Spain, France, and other EU deposit locations. In none of the panels is the effect on the treatment group statistically significant. While we can observe a reduction in deposits located in Spain (a) and an increase in deposits located in France (b), the results are not statistically significant. The findings, therefore, corroborate that indeed DAC6, instead of other unobservable shocks occurring at the same time, is accountable for the movements in deposits.

Figure 6: Dynamic effect of DAC6 on EU-owned deposits: placebo



(a) Deposits in Spain

(b) Deposits in France

Note: the figure shows the coefficients, each of which marks the change in cross-border deposits held in (a) Spain and (b) France around the DAC6 event date (in event time). The sample is restricted to residents in Ireland, Portugal, and the UK, and OECD countries which are not EU member states. The plotted coefficients are those of the interactions of the EU resident indicator and eight separate indicator variables, each marking one quarter over the sample period relative to the quarter before the DAC6 treatment event date (t = 0). We bin the treatment indicators before t - 4 and after t + 4 at the endpoints and omit the indicator for the period t - 1, which therefore serves as a benchmark and has a coefficient value of 0 (and no confidence interval). The figure plots the coefficient estimates together with their 95 per cent confidence intervals for the DAC6 event date. The dependent variable is the log of cross-border deposits. Ordered country-pair and quarter–year fixed effects are included in Panels (a) and (b). Standard errors are clustered at residence country level.

Source: authors' compilation based on BIS-LBS data.

²³ For more details, see Portugal Ministry of Finance (2008), UK Ministry of Finance (2004), and Revenue—Irish Tax and Customs (2019), respectively.

5.2 Circumventing the DAC6 reporting requirement via CBI/RBI programmes

Main analysis

In this section we test hypothesis 3 by analysing how cross-border deposits in non-EU countries that originate from CBI/RBI countries are affected by the implementation of DAC6. CBI/RBI programmes allow individuals to escape information transmission to the true country of residence and therefore enable them to continue the evasion of taxes. Because DAC6 mandates the reporting of cross-border transactions with an EU nexus (i.e. also those conducted by non-EU residents in the EU), we expect cross-border deposits from CBI/RBI residents to increase only outside the EU.

Table 5 presents the results of the DiD regression model similar to equation (1). The difference to the previous analysis is that the treatment group now constitutes residents of countries offering CBI/RBI programmes. Importantly, this also applies to the two EU countries offering CBI/RBI programmes— Cyprus and Malta—that are included in the treatment group.²⁴

	Cross-border deposits						
	(1)	(2)	(3)	(4)	(5)		
Sample	In	Outside	Non-EU	Non-EU	Non-EU		
	EU	EU	tax havens	non-tax havens	non-tax havens		
				(incl US)	(excl. US)		
Post-DAC6 \times RBICBI residents	-0.029	0.266**	0.242**	0.293*	0.283		
	(0.085)	(0.111)	(0.109)	(0.158)	(0.177)		
Constant	2.163***	2.149***	2.444***	1.831***	1.186***		
	(0.003)	(0.004)	(0.004)	(0.006)	(0.006)		
Observations	18,756	20,340	10,560	9,780	8,472		
R-squared	0.975	0.968	0.975	0.961	0.947		
Country-pair FE	YES	YES	YES	YES	YES		
Deposit-quarter-year	YES	YES	YES	YES	YES		
Clustering			Residen	t country			

Table 5: The effect of DAC 6 on deposits owned by CBI and RBI residents

Note: the table reports the main DiD estimates. The dependent variable is the log of cross-border deposits held by residents of country i in banks in deposit country j at the end of quarter q. The sample period is 2017Q1 to 2019Q4 and the sample excludes residents of EU member states, with the exception of those from Malta and Cyprus. In Column (1) deposit locations consist exclusively of EU countries, while in Column (2) they are of non-EU countries. In Column (3), Chile, Guernsey, Hong Kong, Isle of Man, Jersey, Macau, and Switzerland are included. In Column (4), Australia, Brazil, Canada, Korea, Mexico, Philippines, South Africa, Taiwan, and the United States are included. In Column (5), all the countries in Column 4 are included except the United States. Ordered country-pair and deposit country × quarter-year fixed effects are included. Standard errors are clustered at residence country level and are reported in parentheses, *** p < 0.01, ** p < 0.05, * p < 0.1.

Source: authors' compilation based on BIS-LBS data.

In Column (1) we present the results for the deposits that are situated in EU deposit locations. We conduct this test to verify whether residents from countries offering CBI/RBI programmes invested in the EU post-DAC6. As expected, the coefficient of the interaction term is negligible and statistically insignificant. Column (2) shows the results for the sub-sample of non-EU deposit locations. The coefficient on the interaction term is positive (30 per cent) and statistically significant, indicating the strong use of these programmes post-DAC6 to circumvent the new disclosure rule. In a given quarter-year, the average amount of deposits held by CBI/RBI residents in our sample of non-EU countries is US\$47 billion. Given our estimates, that amount is increased by US\$14 billion post-DAC6. The different effects between EU and non-EU deposits therefore indicate that CBI/RBI programmes have not become lucrative in general, but are explained by the implementation of DAC6.

²⁴ We restrict the control group to non-EU countries (including non-OECD countries as well) because our analysis in the previous section illustrates that deposits in EU countries reacted to the introduction of DAC6. Hence, adding these countries to the control group would bias the results.

In Columns (3), (4), and (5) we present the results by splitting the non-EU deposit locations into non-EU tax haven locations and non-EU non-tax haven locations, respectively. We perform this test to verify whether the deposits from CBI/RBI countries were primarily moved to tax havens due to the lower tax rates offered by the tax haven deposit locations. The difference between these two analyses emerges from the unclear role of the United States in terms of secrecy. Given that the United States is not participating in the CRS, the country could arguably be treated as a haven.²⁵ However, so far, there is no general consensus around the status of the United States as a tax haven.²⁶ For this reason, we label the United States as a non-haven country in Column (4) and we exclude it from the list of non-haven countries in Column (5) as an additional test.

When looking at the effect of DAC6 on deposits of CBI country residents, we observe that deposits significantly increase (27 per cent) in tax haven countries, which translates into an approximately. US\$7 billion increase considering an average quarter-year cross-border deposit volume of approximately US\$28 billion in tax havens from CBI/RBI residents. Looking at how deposits in non-haven countries are affected by DAC6, we find that the result is sensitive to the inclusion of the United States as a haven country. While DAC6 significantly increases (34 per cent) the deposits in non-haven countries when we include the United States in the list of non-haven countries (Column (4)), this effect becomes statistically insignificant when we drop the United States from the sample (Column (5)), though the coefficient and the standard deviation are quite comparable.

The latter result hints at the potentially special role of the United States for tax evasion and may explain why EU residents' deposits in the United States remain unaffected by the introduction of DAC6 (cf. Table 4, Column (3)). Instead of owning deposits directly in the United States, EU residents may own them indirectly through shell companies (trust) in a tax haven, such as the Bahamas. This evasion strategy is important to understand the difference between the CRS and DAC6. Under the CRS, trusts are considered non-reportable institutions. However, under DAC6, any transaction from an EU country to the tax haven, in our example the Bahamas, which is done to circumvent the CRS has to be reported, not by the individual itself, but by the intermediary. To circumvent the reporting duty under DAC6, individuals can set up the trust using CBI/RBI citizenship rights instead of directly from the EU country because information only flows to the stated residence country, which in this case is the CBI/RBI country. The consequence of this strategy is that deposits in tax havens are no longer owned by EU residents, but by residents of CBI/RBI countries. Although we are unable to test whether deposits of EU residents decrease in tax havens due to data availability, our results show that deposits in tax havens owned by residents of CBI/RBI countries significantly increase after DAC6. Although tax havens are the natural place to expect deposits to flow, we also observe an increase in deposits in non-tax havens. This result is in line with recent evidence that rich individuals hide their wealth in the form of tangible assets, such as yachts and houses, in high-tax countries like Norway and the United States (see Harding 2022; Lambert 2022). While our results are indirect by nature, they are consistent with a change in the tax evasion strategy induced by DAC6.

Figure 7 shows the graphical results of the event study for the treatment group of CBI/RBI countries in EU and non-EU deposit locations. Panel (a) presents the results for deposits in non-EU deposit locations and shows an immediately significant effect that is constantly increasing in the post-treatment period. Panel (b) presents the results of the deposits in EU countries and indicates that the effect of CBI/RBI programmes is insignificant throughout the entire period of the study. This suggests that the residents of CBI/RBI countries avoided the EU deposit locations post-DAC6.

²⁵ See, for example, Casi et al. (2020), who highlight the special role of the United States in the context of the CRS.

²⁶The United States is not on the EU blacklist or the OECD list for non-cooperative countries.

Figure 7: Dynamic effect of DAC6 on deposits owned by CBI and RBI residents



Note: the figure shows the coefficients, each of which marks the change in cross-border deposits held (a) in the EU and (b) outside the EU. The sample excludes residents of EU member states, with the exception of those from Malta and Cyprus. The plotted coefficients are those of the interactions of the EU resident indicator and eight separate indicator variables, each marking one quarter over the sample period relative to the quarter before the DAC6 treatment event date (t = 0). We bin the treatment indicators before t - 4 and after t + 4 at the endpoints and omit the indicator for period t - 1, which therefore serves as a benchmark and has a coefficient value of 0 (and no confidence interval). The figure plots the coefficient estimates together with their 95 per cent confidence intervals for the DAC6 event date. Ordered country-pair and deposit country × quarter–year fixed effects are included in all panels. Standard errors are clustered at residence country level. Source: authors' compilation based on BIS-LBS data.

Figure 8 shows the results of the event study design in various sub-samples of the data. In Panel (a), we observe a steady increase in deposits in non-EU tax haven locations, which becomes significant in the final period. Panel (b) shows an immediately significant increase in deposits in non-EU non-tax haven locations that remains significant for most of the post-treatment periods. Panel (c) presents the results for the sub-sample of deposits in non-EU non-tax haven locations excluding the United States and shows a very similar picture to Panel (b), with the only difference that the effect becomes statistically insignificant in the last period.²⁷

The use of CBI/RBI programmes in and outside the EU

Figure 9 shows the event study results for the deposits owned by CBI/RBI residents. Panel (a) shows the results for non-EU CBI/RBI countries as the treatment group and non-EU countries as the control group. Here, we observe a strong upward trend in the post-treatment period. The effect becomes significant in the period t = 1 and remains significant for most of the remaining post-treatment periods. In Panel (b) we verify the same results as that of Panel (a), but this time the treatment group comprises only the EU CBI/RBI countries, namely Malta and Cyprus. From Panel (b) we observe that there is a constant upward trend in the post-treatment periods. The effect becomes significant in the period t = 2 and remains significant for the remaining post-treatment periods. The results from this additional test show that there is a strong affinity for the CBI/RBI programmes post-DAC6, whether the country is situated in the EU or outside.

²⁷ In Appendix A3 we show the results from the same analysis but we split the results into EU and non-EU CBI and RBI countries. The results suggest that the use of CBI and RBI programmes extends to those countries within the EU, namely Malta and Cyprus. One possible explanation is that tax evaders might perceive no risk of detection if the transactions are reported to such countries and not their true country of residence.



Figure 8: Dynamic effect of DAC6 on deposits owned by CBI and RBI residents



(b) Deposits in non-EU non-tax havens



(c) Deposits in non-EU non-tax havens excluding the United States

Note: the figure shows the coefficients, each of which marks the change in cross-border deposits held in (a) Chile, Guernsey, Hong Kong, Isle of Man, Jersey, Macau, Switzerland, (b) Australia, Brazil, Canada, Korea, Mexico, Philippines, South Africa, Taiwan, and the United States, and (c) Australia, Brazil, Canada, Korea, Mexico, Philippines, South Africa, and Taiwan around the DAC6 event date (in event time). The sample excludes residents of EU member states, with the exception of those from Malta and Cyprus. The plotted coefficients are those of the interactions of the EU resident indicator and eight separate indicator variables, each marking one quarter over the sample period relative to the quarter before the DAC6 treatment event date (t = 0). We bin the treatment indicators before t - 4 and after t + 4 at the endpoints and omit the indicator for period t - 1, which therefore serves as a benchmark and has a coefficient value of 0 (and no confidence interval). The figure plots the coefficient estimates together with their 95 per cent confidence intervals for the DAC6 event date. Ordered country-pair and deposit country \times quarter–year fixed effects are included in all panels. Standard errors are clustered at residence country level. Source: authors' compilation based on BIS-LBS data.







(b) Including only residents of Malta and Cyprus

Note: the figure shows the coefficients, each of which marks the change in cross-border deposits held outside the EU (a) excluding those held by residents of Malta and Cyprus and (b) including only those held by residents of Malta and Cyprus around the DAC6 event date (in event time). The sample includes residents of Malta and Cyprus and residents of countries that are not in the EU and have no CBI or RBI programmes. The plotted coefficients are those of the interactions of the EU resident indicator and eight separate indicator variables, each marking one quarter over the sample period relative to the quarter before the DAC6 treatment event date (t = 0). We bin the treatment indicators before t - 4 and after t + 4 at the endpoints and omit the indicator for period t - 1, which therefore serves as a benchmark and has a coefficient value of 0 (and no confidence interval). The figure plots the coefficient estimates together with their 95 per cent confidence intervals for the DAC6 event date. Ordered country-pair and deposit country × quarter–year fixed effects are included in both panels. Standard errors are clustered at residence country level.

Source: authors' compilation based on BIS-LBS data.

6 Robustness checks

In this section we briefly discuss the tests we conduct to check the robustness of our main findings. First, we conduct a split sample analysis to rule out that our main findings are driven by the control group. We test the changes in cross-border deposits on the sub-sample of EU residents and non-EU residents separately. Figure A1 presents the results for a time trend test of cross-border deposits in different EU locations. When considering the effect of DAC6 in different EU locations together and separately, as well as outside the EU, the figure shows that the control group is unaffected by the introduction of DAC6. Figure A2 confirms this robustness also for our analysis on CBI/RBI countries.

Second, we examine the reliance of our identification on the fixed effects we select. Specifically, we modify the fixed effects structure of each main result of our analysis. In Figures A3–A5 we include only country-pair fixed effects, only deposit country–time fixed effects, and no fixed effects. Including or excluding fixed effects does not affect our results substantially. The necessity to control for deposit country–time variant characteristics (or for general time trends) is limited given the selection of a short sample period that ensures that no major economic shocks affect cross-border deposits. The necessity to control for time-invariant characteristics across and within country-pairs is also restricted given the accurate selection of the control and treated group in terms of comparability.

Third, in our main test we restrict the sample to the period from the first quarter of 2017 to the last quarter of 2019 in order to exclude possible confounding impacts of the introduction of FATCA and the CRS in 2010–16 as well as the global pandemic in 2020. Nevertheless, in the Appendix we test how our results change if we extend our sample period from the first quarter of 2016 to the last quarter of 2020. In Figure A6 we show that the effect of DAC6 on cross-border deposits in Spain and France continues to hold over the whole sample period. The results in Figure A7 suggest that residents of CBI/RBI countries increase cross-border deposits in tax havens immediately after DAC6 is introduced, although the effect

diminishes in the last quarters of 2020. Instead, the increase in cross-border deposits in non-tax havens outside the EU persists throughout the sample period.

7 Conclusions

This paper studies the effect of the MDR introduced by the EU under the sixth amendment to the Directive on Administrative Cooperation, the so-called DAC6, which came into effect in June 2018. By making intermediaries liable to report a comprehensive set of information on aggressive tax structures, the directive adds a new dimension to the disclosure of cross-border financial activity. In the first part of the study, using data on cross-border deposits, we find that the reporting of cross-border deposits from EU residents in EU deposit locations has considerably increased, with no effect of DAC6 on non-EU deposit locations. However, this result neglects important cross-country differences within the EU. While DAC6 had a negative effect on deposit growth in countries with a stronger enforcement of DAC6, deposits increased in countries with a weaker enforcement of DAC6. These results suggest that individuals partially circumvented the DAC6 disclosure requirement by relocating deposits to weak enforcement countries that offer professional legal privileges for financial institutions.

In the second part of the study, we investigate the effect of CBI/RBI schemes and their usage as methods of regulatory arbitrage against DAC6. We show that cross-border deposits of residents of CBI/RBI countries in the EU were unaffected by DAC6, demonstrating that pre-DAC6 such schemes were not widely used as methods to escape the AEOI. Yet, CBI/RBI country residents sharply increased their deposits in non-EU locations, providing evidence that CBI/RBI schemes can be used to escape the disclosure requirement under DAC6.

Overall, our study contributes substantially to the current international debate on increasing regulation of tax advisory services. A key finding is that strong enforcement is an essential element to increased information collection and exchange on cross-border transactions and to ensuring a disclosure reaction from tax evaders. Moreover, we provide novel evidence on the regulatory arbitrage offered by the existence of risky CBI/RBI programmes.

References

- Allingham, M., and A. Sandmo (1972). 'Income Tax Evasion: A Theoretical Analysis'. Journal of Public Economics, 1: 127–40. https://doi.org/10.1016/0047-2727(72)90010-2
- Almunia, M., and D. Lopez-Rodriguez (2018). 'Under the Radar: The Effects of Monitoring Firms on Tax Compliance'. *American Economic Journal: Economic Policy*, 10(1): 1–38. https://doi.org/10.1257/pol.20160229
- Alstadsæter, A., N. Johannesen, and G. Zucman (2018). 'Who Owns the Wealth in Tax Havens? Macro Evidence and Implications for Global Inequality'. *Journal of Public Economics*, 162: 89–100. https://doi.org/10.1016/j. jpubeco.2018.01.008
- BIS (2020). 'Global Coverage of BIS Locational Banking Statistics'. Report. Available at: https://www.bis.org/ statistics/lbs_globalcoverage.pdf (accessed 11 May 2020).
- Bomare, J., and S.L.G. Herry (2022). 'Will We Ever Be Able to Track Offshore Wealth? Evidence from the Offshore Real Estate Market in the UK'. Working Paper 4. Paris: EU Tax Observatory.
- Carrillo, P., D. Pomeranz, and M. Singhal (2017). 'Dodging the Taxman: Firm Misreporting and Limits to Tax Enforcement'. *American Economic Journal: Applied Economics*, 9(2): 144–64. https://doi.org/10.1257/app. 20140495

- Casi, E., C. Spengel, and B. Stage (2020). 'Cross-Border Tax Evasion after the Common Reporting Standard: Game Over?' *Journal of Public Economics*, 190: 104240. https://doi.org/10.1016/j.jpubeco.2020.104240
- Casi, E., X. Chen, M.D. Orlic, and C. Spengel (2021). 'One Directive, Several Transpositions: A Cross-Country Evaluation of the National Implementation of DAC6'. *World Tax Journal*, 13(1): 63–81.
- Christians, A. (2017). 'Buying In: Residence and Citizenship by Investment'. *St. Louis University Law Journal*, 62: 51.
- De Simone, L., R. Lester, and K. Markle (2020). 'Transparency and Tax Evasion: Evidence from the Foreign Account Tax Compliance Act (FATCA)'. *Journal of Accounting Research*, 58(1): 105–53. https://doi.org/10. 1111/1475-679X.12293
- Devereux, M., J. Freedman, and J. Vella (2012). 'Review of DOTAS and the Tax Avoidance Landscape: Paper 2'. CBT Report. Oxford: Oxford University Centre for Business Taxation.
- Edwards, A., M. Hutchens, and A. Persson (2021). 'Do Third-Party Cross-Border Tax Transparency Requirements Impact Firm Behavior?' Working Paper 3792342. Toronto: Rotman School of Management. https://doi.org/10. 2139/ssrn.3792342
- European Commission (2017). 'Commission Staff Working Document Executive Summary of the Impact Assessment Accompanying the Document Proposal for a Council Directive Amending Directive 2011/16/EU as Regards Mandatory Automatic Exchange of Information in the Field of Taxation in Relation to Reportable Cross-Border Arrangements'. SWD/2017/0237 final 2017/0138 (CNS). Available at: https://eur-lex.europa.eu/legal-content/EN/ALL/?uri=SWD%3A2017%3A0237%3AFIN (accessed 13 July 2022).
- European Council (2011). 'Council Directive 2011/16/EU of 15 February 2011 on Administrative Cooperation in the Field of Taxation and Repealing Directive' 77/799/eec. *Official Journal of the European Union*, 64: 1–12.
- European Council (2017). 'Proposal for a Council Directive Amending Directive 2011/16/EU as Regards Mandatory Automatic Exchange of Information in the Field of Taxation in Relation to Reportable Cross-Border Arrangements'. COM/2017/0335 final. Available at: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri= CELEX%3A52017PC0335 (accessed 12 September 2022).
- European Council (2018). 'Council Directive (EU) 2018/822 of 25 May 2018 Amending Directive 2011/16/EU as Regards Mandatory Automatic Exchange of Information in the Field of Taxation in Relation to Reportable Cross-Border Arrangements'. *Official Journal of the European Union*, 139: 1–13.
- European Parliament (2016). 'European Parliament Resolution of 6 July 2016 on Tax Rulings and Other Measures Similar in Nature or Effect (2016/2038(INI))'. Available at: http://www.europarl.europa.eu/doceo/document/ TA-8-2016-0310_EN.pdf (accessed 12 September 2022).
- European Parliament (2018). 'Citizenship by Investment (CBI) and Residency by Investment (RBI) Schemes in the EU'. Study. Brussels: European Parliamentary Research Service.
- EY (2020). 'France Publishes Additional Official Tax Guidelines on Mandatory Disclosure Rules'. Available at: https://taxnews.ey.com/news/2020-1283-france-publishes-additional-official-tax-guidelines-onmandatory-disclosure-rules (accessed 12 September 2022).
- Fack, G., and C. Landais (2010). 'Are Tax Incentives for Charitable Giving Efficient? Evidence from France'. *American Economic Journal: Economic Policy*, 2(2): 117–41. https://doi.org/10.1257/pol.2.2.117
- Fuest, C., A. Peichl, and S. Siegloch (2018). 'Do Higher Corporate Taxes Reduce Wages? Micro Evidence from Germany'. American Economic Review, 108(2): 393–418. https://doi.org/10.1257/aer.20130570
- Guyton, J., P. Langetieg, D. Reck, M. Risch, and G. Zucman (2021). 'Tax Evasion at the Top of the Income Distribution: Theory and Evidence'. Working Paper 28542. Cambridge, MA: National Bureau of Economic Research. https://doi.org/10.3386/w28542
- Harding, L. (2022). 'Russian Oligarch's Yacht Stuck in Norway as Suppliers Refuse to Refuel'. *Guardian*. Available at: https://www.theguardian.com/world/2022/mar/18/russian-oligarchs-yacht-stuck-in-norway-as-suppliers-refuse-to-refuel (accessed 14 June 2022).

- Haines, A. (2020). 'Taxpayers Moving Away from the Big Four to Law Firms for Advice'. International Tax Review. Available at: https://www.internationaltaxreview.com/article/2a6a4ynkdzu2frfmacjk0/taxpayersmoving-away-from-the-big-four-to-law-firms-for-advice (accessed 12 September 2022).
- Huizinga, H., and G. Nicodéme (2004). 'Are International Deposits Tax-Driven'. *Journal of Public Economics*, 88(6): 1093–118. https://doi.org/10.1016/S0047-2727(03)00058-6
- International Monetary Fund (2015). 'Current Challenges in Revenue Mobilization: Improving Tax Compliance'. Policy Paper. Washington, DC: IMF. https://doi.org/10.5089/9781498344890.007
- Johannesen, N., and G. Zucman (2014). 'The End of Bank Secrecy? An Evaluation of the G20 Tax Haven Crackdown'. *American Economic Journal: Economic Policy*, 6(1): 65–91. https://doi.org/10.1257/pol.6.1.65
- Johannesen, N., P. Langetieg, D. Reck, M. Risch, and J. Slemrod (2020). 'Taxing Hidden Wealth: The Consequences of US Enforcement Initiatives on Evasive Foreign Accounts'. *American Economic Journal: Economic Policy*, 12(3): 312–46. https://doi.org/10.1257/pol.20180410
- Joshi, P. (2020). 'Does Private Country-by-Country Reporting Deter Tax Avoidance and Income Shifting? Evidence from BEPS Action Item 13'. *Journal of Accounting Research*, 58(2): 333–81. https://doi.org/10.1111/ 1475-679X.12304
- Kleven, H.J., M.B. Knudsen, C.T. Kreiner, S. Pedersen, and E. Saez (2011). 'Unwilling or Unable to Cheat? Evidence from a Tax Audit Experiment in Denmark'. *Econometrica*, 79(3): 651–92. https://doi.org/10.3982/ ECTA9113
- Kopczuk, W., J. Marion, E. Muehlegger, and J. Slemrod (2016). 'Does Tax-Collection Invariance Hold? Evasion and the Pass-Through of State Diesel Taxes'. *American Economic Journal: Economic Policy*, 8(2): 251–86. https://doi.org/10.1257/pol.20140271
- Lambert, L. (2022). "They're Scared": Wealthy Russians Look to Sell U.S. Real Estate Everywhere from Fisher Island to Billionaires' Row'. *Fortune*. Available at: https://fortune.com/2022/03/04/wealthy-russians-sellingamerican-real-estate-amid-ukraine-crisis/ (accessed 14 June 2022).
- Langenmayr, D., and L. Zyska (2021). 'Escaping the Exchange of Information: Tax Evasion via Citizenship-by-Investment'. Working Paper 8956. Munich: CESifo. https://doi.org/10.2139/ssrn.3812073
- Mehboob, D. (2019). 'Residence-by-Investment Schemes Continue to Create Tax Loopholes'. International Tax Review. Available at: https://www.internationaltaxreview.com/article/b1fyb354wbxtkx/residenceby-investment-schemes-continue-to-create-tax-loopholes (accessed 11 May 2020).
- Menkhoff, L., and J. Miethe (2019). 'Tax Evasion in New Disguise? Examining Tax Havens' International Bank Deposits'. Journal of Public Economics, 176: 53–78. https://doi.org/10.1016/j.jpubeco.2019.06.003
- Miethe, J. (2020). 'The Elusive Banker: Using Hurricanes to Uncover (Non-)Activity in Offshore Financial Centers'. CESifo Working Paper 8625. Munich: CESifo. https://doi.org/10.2139/ssrn.3715277
- Noked, N., and Z. Marcone (2022). 'Targeting Tax Avoidance Enablers'. UC Irvine Law Review.
- Noked, N., Z. Marcone, and A. Tsang (2022). 'The Expansion and Internationalization of Mandatory Disclosure Rules'. Columbia Journal of Tax Law, 13: 122. https://doi.org/10.52214/cjtl.v13i2.9798
- OECD (2013). Action Plan on Base Erosion and Profit Shifting. Paris: OECD Publications.
- OECD (2018). 'Residence/Citizenship by Investment Schemes'. Available at: https://www.oecd.org/tax/ automatic-exchange/crs-implementation-and-assistance/residence-citizenship-by-investment/ (accessed 11 May 2020).
- OECD (2020). 'Compilation of Comments: Public Input Received on Misuse of Residence by Investment Schemes to Circumvent the Common Reporting Standard'. Available at: https://www.oecd.org/tax/exchange-of-tax-information/public-input-received-misuse-of-residence-by-investment-schemes-to-circumvent-the-common-reporting-standard.pdf (accessed 11 May 2020).
- Olbert, M., and L. De Simone (2021). 'Real Effects of Private Country-By-Country Disclosure'. Accounting Review. https://doi.org/10.2308/TAR-2020-0714

- Portugal Ministry of Finance (2008). 'Decree-Law No. 29/2008 of 25 February 2008'. Available at: https://dre. tretas.org/dre/229553/decreto-lei-29-2008-de-25-de-fevereiro (accessed 11 January 2022).
- Revenue—Irish Tax and Customs (2019). 'Guidance Notes on Mandatory Disclosure Regime'. Available at: https://www.revenue.ie/en/tax-professionals/documents/notes-for-guidance/mandatory-disclosure/ guidance-notes-mandatory-disclosure-regime-2015.pdf (accessed 11 January 2022).
- Schmidheiny, K., and S. Siegloch (2019). 'On Event Studies and Distributed-Lags in Two-Way Fixed Effects Models: Identification, Equivalence, and Generalization'. Discussion Paper 20-017. Mannheim: ZEW: Centre for European Economic Research. https://doi.org/10.2139/ssrn.3571164
- Slemrod, J. (2019). 'Tax Compliance and Enforcement'. *Journal of Economic Literature*, 57(4): 904–54. https://doi.org/10.1257/jel.20181437
- UK Ministry of Finance (2004). 'Finance Act 2004, Part 7'. https://www.legislation.gov.uk/ukpga/2004/12/part/7 (accessed 11 January 2022).
- US Senate Finance Committee (2022). 'The Shell Bank Loophole'. Report. Washington, DC: US Senate Finance Committee. Available at: https://www.finance.senate.gov/imo/media/doc/Mirabaud%20Report.pdf (accessed 14 June 2022).
- Zucman, G. (2013). 'The Missing Wealth of Nations: Are Europe and the US Net Debtors or Net Creditors?' *Quarterly Journal of Economics*, 128(3): 1321–64. https://doi.org/10.1093/qje/qjt012

Appendix A

A1 Main results: split tests



Figure A1: Dynamic effect of DAC6 on EU-owned deposits: split test

(e) Deposits outside EU

Note: the figure replicates Figure 5 panels (b), (c), (d) and (e) and Figure 4 panel (b). The figure shows the coefficients, each of which marks the change in cross-border deposits held by residents of EU (red) and non-EU (blue) countries in (a) Spain, (b) France, (c) France and Austria, (d) Denmark, the UK, Italy, and Sweden, and (e) Australia, Canada, Switzerland, the United States, and South Africa around the DAC6 event date (in event time).

Source: authors' compilation based on BIS-LBS data.



Figure A2: Dynamic effect of DAC6 on deposits owned by CBI and RBI residents: split test

(e) Deposits in non-EU non-tax haven (excluding United States)

Control group

-1

-4+

-3

Treated group

-2

Note: the figure replicates Figure 7 panels (a) and (b) and Figure 8 panel (a) and (b). The figure shows the coefficients, each of which marks the change in cross-border deposits held by residents of CBI/RBI countries (red) and non-CBI/RBI countries (blue) in (a) all non-EU deposit locations, (b) all EU deposit locations, (c) Chile, Guernsey, Hong Kong, Isle of Man, Jersey, Macau, and Switzerland, (d) Australia, Brazil, Canada, Korea, Mexico, Philippines, South Africa, Taiwan, and the United States, and (e) in Australia, Brazil, Canada, Korea, Mexico, Philippines, South Africa, and Taiwan around the DAC6 event date (in event time). Source: authors' compilation based on BIS-LBS data.

2

3

4+

A2 Main results: different fixed effects structure



Figure A3: Dynamic effect of DAC6 on deposits in Spain owned by EU residents

(c) No fixed effects

-1 0 1 t= DAC6 introduction 2

3

4+

-2

-3

Note: The figure replicates Figure 5 panel (b) but with different fixed effects. Ordered country-pair fixed effects are included in (a), quarter-year fixed effects are included in (b), no fixed effect is included in (c). Standard errors are clustered at residence country level.

Source: authors' compilation based on BIS-LBS data.

ې۔ ++



Figure A4: Dynamic effect of DAC 6 on deposits in France owned by EU residents



(b) Only quarter-year fixed effects



(c) No fixed effects

Note: the figure replicates Figure 5 panel (c) but with different fixed effects. Ordered country-pair fixed effects are included in (a), quarter-year fixed effects are included in (b), no fixed effect is included in (c). Source: authors' compilation based on BIS-LBS data.





(a) Only country-pair fixed effects

(b) Only deposit country-quarter-year fixed effects



(c) No fixed effects

Note: the figure replicates Figure 7 panels (a) and (b) but with different fixed effects. Ordered country-pair fixed effects are included in (a), deposit country-quarter-year fixed effects are included in (b), no fixed effect is included in (c). Standard errors are clustered at residence country level.

Source: authors' compilation based on BIS-LBS data.

A3 Main results: long-term effect



Figure A6: Dynamic effect of DAC6 on EU-owned foreign deposits: long run

Note: the figure replicates Figure 5 panels (b) and (c). A longer period of time is considered from 2016Q1 to 2020Q4. Source: authors' compilation based on BIS-LBS data.



Figure A7: Dynamic effect of DAC6 on deposits owned by CBI and RBI residents: long run

(a) Deposits in non-EU tax havens

(b) Deposits in non-EU non-tax havens

Note: the figure replicates Figure 9 panels (a) and (b). A longer period of time is considered from 2016Q1 to 2020Q4. Source: authors' compilation based on BIS-LBS data.